Author's response to reviews

Title: School difficulties and role of social, material, behavioural, physical and mental resources among multi-cultural students

Authors:

Kénora Chau (c.kenora@yahoo.fr)
Michèle Baumann (michele.baumann@uni.lu)
Bernard Kabuth (kabuth_b@yahoo.fr)
Nearkasen Chau (Nearkasen.Chau@wanadoo.fr)

Version: 2 Date: 12 January 2012

Author's response to reviews:

We have modified the manuscript as you have proposed.
beste regards
Michèle Baumann